Author's response to reviews

Title: Immunological adjuvant effect of the peptide fraction from the larvae of Musca domestica

Authors:

Liqing Chen (chenjin52611@163.com)
Juan Zhang (zjatzju@126.com)
Hongxiang Sun (sunhx@zju.edu.cn)

Version: 5 Date: 24 November 2015

Author's response to reviews: see over
Dear Dr. Tom Rowles,

Thank you very much for your mail about the decision on our manuscript entitled "Immunological adjuvant effect of the peptide fraction from the larvae of *Musca domestica*" (MS: 7326976121677088). Now we have revised the manuscript according to your suggestion and the Editorial comments.

Editorial comments:

Please include details in your Methods section on who formally identified the *Musca domestica* used in your study.

The sentence “A voucher specimen (No. 20101105) has been deposited at the Laboratory of Nature Drug, College of Animal Sciences, Zhejiang University, China, and identified by professor Jun-An Ye at College of Animal Sciences, Zhejiang University” has been supplemented in the revised manuscript.
We are looking forward to hear from you.

Sincerely yours,

Prof. Dr Hongxiang Sun